## Global and regional marine ecosystem model climate change projections reveal key uncertainties

Tyler D Eddy<sup>1</sup>, Ryan F Heneghan<sup>2</sup>, Andrea Bryndum-Buchholz<sup>1</sup>, Elizabeth A Fulton<sup>3,4</sup>, Cheryl S Harrison<sup>5</sup>, Derek P Tittensor<sup>6</sup>, Heike K Lotze<sup>6</sup>, Kelly Ortega-Cisneros<sup>7,8</sup>, Camilla Novaglio<sup>4,9</sup>, Daniele Bianchi<sup>10</sup>, Matthias Büchner<sup>11</sup>, Catherine Bulman<sup>3</sup>, William WL Cheung<sup>12</sup>, Villy Christensen<sup>12,13</sup>, Marta Coll<sup>13,14</sup>, Jason D Everett<sup>15,16,17</sup>, Denisse Fierro-Arcos<sup>9</sup>, Eric D Galbraith<sup>18</sup>, Didier Gascuel<sup>19</sup>, Jerome Guiet<sup>10</sup>, Steve Mackinson<sup>20</sup>, Olivier Maury<sup>21</sup>, Susa Niiranen<sup>22</sup>, Ricardo Oliveros-Ramos<sup>23</sup>, Juliano Palacios-Abrantes<sup>12</sup>, Chiara Piroddi<sup>24</sup>, Hubert du Pontavice<sup>25</sup>, Jonathan Reum<sup>26</sup>, Anthony J Richardson<sup>15,16</sup>, Jacob Schewe<sup>11</sup>, Lynne Shannon<sup>8</sup>, Yunne-Jai Shin<sup>23</sup>, Jeroen Steenbeek<sup>13</sup>, Jan Volkholz<sup>11</sup>, Nicola D Walker<sup>27</sup>, Phoebe Woodworth-Jefcoats<sup>28</sup>, Julia L Blanchard<sup>4,9</sup>

<sup>1</sup>Centre for Fisheries Ecosystems Research, Fisheries & Marine Institute, Memorial University, St. John's, NL, Canada

<sup>2</sup>Australian Rivers Institute, School of Environment and Science, Griffith University, Nathan, Queensland, Australia

<sup>3</sup>CSIRO Environment, Hobart, Tasmania, Australia

<sup>4</sup>Centre for Marine Socioecology, University of Tasmania, Hobart, Tasmania, Australia

<sup>5</sup>Department of Oceanography & Coastal Sciences, Louisiana State University, Baton Rouge, LA, USA

<sup>6</sup>Department of Biology, Dalhousie University, Halifax, NS, Canada

<sup>7</sup>Institute for Coastal and Marine Research, Nelson Mandela University, Gqeberha, South Africa <sup>8</sup>Department of Biological Sciences, University of Cape Town, Cape Town, South Africa

<sup>9</sup>Institute for Marine and Antarctic Studies, University of Tasmania, Hobart, Tasmania, Australia <sup>10</sup>Department of Atmospheric and Oceanic Sciences, University of California Los Angeles, CA, USA

<sup>11</sup>Potsdam Institute for Climate Impact Research, Member of the Leibniz Association, Potsdam, Germany

<sup>12</sup>Institute for the Oceans and Fisheries, University of British Columbia, Vancouver, BC Canada <sup>13</sup>Ecopath International Initiative, Barcelona, Spain

<sup>14</sup>Institute of Marine Sciences, Barcelona, Spain

<sup>15</sup>School of the Environment, University of Queensland, St Lucia, QLD, Australia

<sup>16</sup>CSIRO Oceans and Atmosphere, Queensland Biosciences Precinct, St Lucia, QLD, Australia

<sup>17</sup>Centre for Marine Science and Innovation, School of Biological, Earth and Environmental

Science, University of New South Wales, Sydney, NSW Australia

<sup>18</sup>Department of Earth and Planetary Science, McGill University, Montreal, QC, Canada <sup>19</sup>Institut Agro, Ifremer, INRAE, Rennes, France

<sup>20</sup>Scottish Pelagic Fishermen's Association, Fraserburgh, Aberdeenshire, UK

<sup>21</sup>MARBEC, IRD, Univ Montpellier, IFREMER, CNRS, Sete, France

<sup>22</sup>Stockholm Resilience Centre, Stockholm University, Sweden

<sup>23</sup>MARBEC, IRD, Univ Montpellier, IFREMER, CNRS, Montpellier, France

<sup>24</sup>European Commission, Joint Research Centre, Ispra, Italy

<sup>25</sup>Ifremer, HMMN, Laboratoire Ressources Halieutiques, Port-en-Bessin, France

<sup>26</sup>NOAA Alaska Fisheries Science Center, Seattle, WA, USA

<sup>27</sup>Centre for Environment, Fisheries and Aquaculture Science (Cefas), Lowestoft Laboratory, Lowestoft, UK

<sup>28</sup> Pacific Islands Fisheries Science Center, National Marine Fisheries Service, National Oceanic and Atmospheric Administration, Honolulu, HI, USA

- 1 Key points
- Global marine ecosystem models projected greater biomass declines with climate
   change than regional marine ecosystem models for many regions
- For both global and regional models, greater biomass declines were projected in CMIP6
   than CMIP5 and in IPSL vs. GFDL simulations
- Projected impacts of climate change on marine ecosystems at regional scales are
   currently less certain than at global scale
- 8
- 9 Abstract

10 Climate change is affecting ocean temperature, acidity, currents, and primary production, causing shifts in species distributions, marine ecosystems, and ultimately fisheries. 11 Earth system models simulate climate change impacts on physical and biogeochemical 12 13 properties of future oceans under varying emissions scenarios. Coupling these simulations with 14 an ensemble of global marine ecosystem models has indicated broad decreases of fish biomass 15 with warming. However, regional details of these impacts remain much more uncertain. Here, 16 we employ CMIP5 and CMIP6 climate change impact projections using two Earth system 17 models coupled with four regional and nine global marine ecosystem models in ten ocean 18 regions to evaluate model agreement at regional scales. We find that models developed at 19 different scales can lead to stark differences in biomass projections. On average, global models 20 projected greater biomass declines by the end of the 21st century than regional models. For 21 both global and regional models, greater biomass declines were projected using CMIP6 than 22 CMIP5 simulations. Global models projected biomass declines in 86% of CMIP5 simulations for 23 ocean regions compared to 50% for regional models in the same ocean regions. In CMIP6 simulations, all global model simulations projected biomass declines in ocean regions by 2100, 24 while regional models projected biomass declines in 67% of the ocean region simulations. Our 25 26 analysis suggests that improved understanding of the causes of differences between global and

27 regional marine ecosystem model climate change projections is needed, alongside28 observational evaluation of modelled responses.

29

30 Plain language summary

31 Climate change is affecting the world's oceans, marine ecosystems, biodiversity, and the 32 ecosystem services that they support, including fisheries that feed millions of people worldwide. 33 Anticipating the impacts of climate change can help society and managers to prepare for, and 34 adapt to, changes ahead. Present understanding of climate change impacts on the world's oceans based on global models indicates a 5% loss in animal biomass with every 1 °C that the 35 planet warms. Here, we compare potential future biomass on regional scales that are most 36 37 relevant for management decisions about sustainable resource use. We used regional scale 38 ecosystem models tailored to the species and fisheries they represent. We compared climate 39 change projections of ocean biomass changes from these regional models to corresponding 40 areas from global models to see how well they agreed. We found key differences in climate 41 change projections of ocean biomass between global and regional models. In some cases, both 42 global and regional models projected biomass declines, while in others global models 43 suggested a decline and regional models an increase. Our study highlights that we need further 44 exploration and understanding of the differences in ocean biomass change between global and 45 regional marine ecosystem models.

46

47

48 Key words

(ISIMIP); fisheries

model intercomparison project (MIP); model ensemble; Fisheries and Marine Ecosystem Model
 Intercomparison Project (FishMIP); Inter-Sectoral Impact Model Intercomparison Project

52

53 Introduction

Oceans play a key role in regulating global climate (IPCC 2023). Marine ecosystems and biodiversity provide a range of ecosystem services including livelihood opportunities, food provision, coastal protection, and carbon sequestration (IPBES 2019). However, ocean ecosystems, marine biodiversity, and the ecosystem services they provide are compromised by anthropogenic climate change impacting water temperature, hydrodynamics, geochemistry, primary productivity, and species and community dynamics (IPBES 2019; IPCC 2023).

60 The United Nations Intergovernmental Panel on Climate Change (IPCC) assessment 61 reports make use of standardized climate change projections from an ensemble of Earth system models (ESMs) through the coupled model intercomparison project (CMIP; Eyring et al. 2016). 62 ESMs provide projections of many variables important for marine life, including sea surface 63 64 temperature, oxygen, hydrodynamics, sea level, primary production, low trophic level biomass, 65 among many other variables (IPCC 2023). By using an ensemble of ESMs, climate change 66 projections are not dependent on any one model, and variation among model projections can be 67 evaluated to determine the level of uncertainty for a given environmental variable (IPCC 2023). 68 For some variables, such as acidity, ensemble model agreement is very good; while others, 69 such as sea surface temperature and primary production show much less agreement (Bopp et al. 2013; Kwiatkowski et al. 2020; IPCC 2023). Understanding sources of uncertainty in model 70 71 projections is a key research focus for climate and marine ecosystem modellers to help build confidence in climate change and impact projections (Cheung et al. 2016; Payne et al. 2016; 72 73 Eddy 2019).

Taking a similar approach to CMIP, the Fisheries and Marine Ecosystem Model
Intercomparison Project (FishMIP) has developed standardized protocols to run climate change
impact simulations for an ensemble of global and regional marine ecosystem models (MEMs;
Tittensor et al. 2018; Blanchard et al. 2024). FishMIP models use outputs provided by CMIP
ESMs to run climate change scenarios following shared socioeconomic pathways (SSPs) and

79 representative concentration pathways (RCPs). Global FishMIP projections have been used to 80 explore changes in marine animal biomass, evaluate the level of MEM agreement, and attribute sources of variability to MEMs, ESMs, SSP-RCPs, and fishing exploitation (Lotze et al. 2019). 81 82 FishMIP projections suggest that marine animal biomass will, on average, decline by 17% by 83 2100 under a high emissions scenario (RCP 8.5) with CMIP5 forcing data (Lotze et al. 2019) 84 and by 19% under CMIP6 (Tittensor et al. 2021). Other key findings are that with every 1 °C that 85 the planet warms, marine animal biomass is projected to decline by 5%, roughly in agreement with the average of observational estimates (Free et al., 2019), and that higher trophic levels are 86 disproportionately impacted (Lotze et al. 2019; du Pontavice et al., 2021; Guibourd de Luzinais 87 et al. 2023). While not all FishMIP models represent fishing, simulations with and without fishing 88 produced similar magnitude and variability of the climate effect on marine ecosystems (Lotze et 89 90 al. 2019). Equivalent amounts of variability in biomass projections were contributed by ESMs 91 and MEMs, with variability increasing with higher emissions (Lotze et al. 2019). Compared to 92 CMIP5 forced MEMs, CMIP6 forced MEMs projected regional differences in the direction of 93 biomass changes, emphasizing the need to reduce uncertainty to support adaptation planning 94 (Tittensor et al. 2021). Understanding regional marine ecosystem and fisheries dynamics is 95 particularly important because many coastal communities are highly dependent on the nutrition 96 and livelihoods they derive from the ocean (e.g. FAO 2018).

97 Regional MEMs have often been developed to address fisheries, conservation, and 98 management applications. In contrast to global models, regional models do not represent the 99 entire ocean, and are generally forced by environmental variables representative of the regions 100 being simulated. This is one of several differences in model detail between global and regional scale MEMs that has consequences for using global scale ESMs to force regional MEMs. One 101 102 particularly important issue is that ESMs produce projections of physical and biogeochemical ocean properties at a coarse spatial resolution, typically on a 1 ° x 1 ° grid. A consequence of 103 104 this spatial resolution is that physical and biogeochemical processes that drive primary

105 productivity in coastal regions – where many nationally significant fisheries occur – are not well 106 represented (FAO 2022). For example, depths <50 m and key oceanographic processes that 107 affect primary production, such as upwelling and dispersion of coastal nutrients, are poorly 108 resolved (Stock et al. 2011; Bopp et al. 2013). This has implications for representation of 109 marginal seas, such as the Baltic Sea (Niiranen et al. 2013) and the Mediterranean Sea (Coll et 110 al. 2010), and shallow underwater plateaus such as The Grand Banks of Newfoundland 111 (Laurent et al. 2021). In Australia, discrepancies in projections from global and regional MEMs 112 for the same regions have been observed, not just in terms of the magnitude of effects but also in the direction of change (Pethybridge et al. 2020). As there is a need to provide projections of 113 climate change impacts at regional scales for fisheries adaptation and mitigation planning. 114 regional MEM ensembles that allow quantification of across-model uncertainty are lacking for 115 116 most regions of the world (Metcalfe et al. 2015). In this absence, the FishMIP global ensemble 117 has been used to fill in the gap (Blanchard et al. 2017; Cinner et al 2022; Blanchard and 118 Novaglio 2024). Understanding how global marine ecosystem models perform at regional 119 scales is important as many resource constrained jurisdictions do not have the capacity to 120 generate their own regional specific models (Barange et al. 2014; Blanchard et al. 2017; Boyce 121 et al. 2022; Cinner et al. 2022).

122 Most global FishMIP models agree in the projected direction of change in fish production 123 for specific climate model and emissions scenarios, albeit with regional differences (Lotze et al. 124 2019; Heneghan et al. 2021; Tittensor et al. 2021). However, models at regional scales can 125 project the opposite direction of change (Barange et al. 2014; Pethybridge et al. 2020). Previous 126 FishMIP studies have analyzed variability in climate change projections of marine animal 127 biomass as a function of: ESM (Geophysical Fluid Dynamics Laboratory Earth System Model -128 GFDL or Institute Pierre Simon Laplace Climate Model – IPSL); global MEM (ensemble of six or 129 nine; Lotze et al. 2019; Tittensor et al. 2021); and SSP-RCP scenario (four scenarios or two scenarios; Lotze et al. 2019; Tittensor et al. 2021). The impact of fishing (Lotze et al. 2019) and 130

131 the difference between CMIP5 and CMIP6 projections have also been investigated (Tittensor et 132 al. 2021). What remains unknown is how climate change projections by FishMIP global MEMs compare with regional MEMs at regional scales. In this study, we explore variability in climate 133 134 change projections of ocean biomass as a function of: Earth system model (ESM; GFDL vs. 135 IPSL); marine ecosystem model (MEM; ensemble of 13); marine ecosystem model spatial scale (global or regional); and couple model intercomparison project generation (CMIP5 vs. CMIP6) in 136 137 10 regions. We evaluate the agreement in projections of global MEMs with regional MEMs at 138 regional scales.

139

140 Methods

141 FishMIP model ensemble

142 FishMIP provides a standardized approach to compare climate change projections 143 among MEMs of varying structure, assumptions, and spatial scale (Tittensor et al. 2018; Eddy 144 2019; Heneghan et al. 2021; Novaglio et al. 2024; Blanchard et al. 2024). MEMs differ in how they are structured – whether biomass or carbon based, structured by size, trophic level, 145 146 species or functional group (Tittensor et al. 2018). Global MEMs are based on general ecological theory and principles, are spatially resolved, and have become a major modelling 147 focus within the last decade (Table 1). Regional models generally represent more trophic 148 149 interactions, are fit to local ecological survey and fisheries data, may or may not be spatially resolved, and have had a long history of development and refinement, originating 40 years ago 150 151 (Polovina 1984). Regional models generally include more feedback processes and the system 152 specific ecological idiosyncrasies that add complexity and richness to regional ecosystems around the globe (Table 1). 153

154

155 Global & regional MEMs

156 We analyzed regional MEM projections for seven regions in CMIP5 and five regions in 157 CMIP6, with 10 regions represented in total (two regions with both CMIP5 and CMIP6 simulations; Figure 1). Regions were selected based on regional models that participated in 158 159 FishMIP simulation protocols (Tittensor et al. 2018; fishmip.org). Global ESMs and MEMs are 160 often not well resolved in coastal regions compared to regional models and therefore even 161 though the spatial domains used were identical, the ecosystems represented may differ (Figure 162 1; Tables 1, 2). For this reason, not all regions included in this study have all global MEM 163 simulations.

An important difference between global and regional model development is that regional 164 models are often forced with one or more time series of: fishing mortality or effort by species or 165 fishery, oceanographic flows that capture current patterns, net primary productivity (NPP), 166 167 temperature, salinity, pH, nutrient and other inflow from coastal sources, and are fit to 168 independent, historical observational data such as fisheries catch data and/or biomass survey 169 data for individual species (Table 1). These observational datasets are often not available at 170 the global scale, which limits the calibration process for global MEMs. As ESM hindcasts do not 171 always match observations at regional scales, all regional MEMs employed bias correction using the delta method to maintain calibration (Schoeman et al. 2023; Table 2) or statistical or 172 dynamical downscaling of ESM outputs (Oliveros-Ramos et al. in revision, Coll et al., 2024). 173 174 This introduced variation in how global and regional models performed climate change simulations, affecting the magnitude of projected biomass. For this reason, we report relative 175 176 changes in biomass.

177

178 FishMIP simulation protocol

179 FishMIP simulations followed a standardized protocol that used ESM output variables as

180 MEM input variables and the same fisheries scenarios to simulate climate change (Frieler et al.

181	2017; Tittensor et al. 2018; Frieler et al. 2024; Blanchard et al. 2024; Figure S1; Tables 1, 2).
182	FishMIP simulations were designed to align with the Inter-Sectoral Impact Model
183	Intercomparison Project (ISIMIP) 2b and 3b protocols corresponding to CMIP5 and CMIP6
184	simulation rounds, respectively (Frieler et al. 2018; Frieler et al. 2024). In both protocols, MEM
185	simulations were run from 1950 – 2100 using climate and oceanographic conditions from two
186	ESMs (GFDL-ESM2M for CMIP5 and GFDL-ESM4 for CMIP6, collectively referred to as GFDL;
187	and IPSL-CM5A-LR for CMIP5 and IPSL-CM6A-LR for CMIP6, collectively referred to as
188	IPSL), under four emissions scenarios (RCP2.6/SSP1-2.6, RCP4.5/SSP2-4.5, RCP6.0/SSP4-
189	6.0, RCP8.5/SSP5-8.5; Tittensor et al. 2018; Frieler et al. 2024; Blanchard et al. 2024).
190	For this analysis, we used no-fishing simulations as most global models do not represent
191	
	fishing and did not run fishing simulations. We focussed on the RCP8.5/SSP5-8.5 high
192	emissions scenario as the impacts of emission scenario have been previously explored for
192 193	emissions scenario as the impacts of emission scenario have been previously explored for CMIP5 and CMIP6 (Lotze et al. 2019; Tittensor et al. 2021). While considering a broader set of
192 193 194	emissions scenario as the impacts of emission scenario have been previously explored for CMIP5 and CMIP6 (Lotze et al. 2019; Tittensor et al. 2021). While considering a broader set of scenarios would be ideal, it was not possible in this instance as only the RCP8.5/SSP5-8.5 had
192 193 194 195	tishing and did not run fishing simulations. We focussed on the RCP8.5/SSP5-8.5 high emissions scenario as the impacts of emission scenario have been previously explored for CMIP5 and CMIP6 (Lotze et al. 2019; Tittensor et al. 2021). While considering a broader set of scenarios would be ideal, it was not possible in this instance as only the RCP8.5/SSP5-8.5 had been run by all models. Extending to another scenario was unfortunately not feasible due to
192 193 194 195 196	tishing and did not run fishing simulations. We focussed on the RCP8.5/SSP5-8.5 high emissions scenario as the impacts of emission scenario have been previously explored for CMIP5 and CMIP6 (Lotze et al. 2019; Tittensor et al. 2021). While considering a broader set of scenarios would be ideal, it was not possible in this instance as only the RCP8.5/SSP5-8.5 had been run by all models. Extending to another scenario was unfortunately not feasible due to computational intensity and capacity as FishMIP and ISIMIP are largely volunteer contributions
192 193 194 195 196 197	tishing and did not run fishing simulations. We focussed on the RCP8.5/SSP5-8.5 high emissions scenario as the impacts of emission scenario have been previously explored for CMIP5 and CMIP6 (Lotze et al. 2019; Tittensor et al. 2021). While considering a broader set of scenarios would be ideal, it was not possible in this instance as only the RCP8.5/SSP5-8.5 had been run by all models. Extending to another scenario was unfortunately not feasible due to computational intensity and capacity as FishMIP and ISIMIP are largely volunteer contributions unlike CMIP. The high emissions scenario samples a large range of global warming, and as
192 193 194 195 196 197 198	tishing and did not run fishing simulations. We focussed on the RCP8.5/SSP5-8.5 high emissions scenario as the impacts of emission scenario have been previously explored for CMIP5 and CMIP6 (Lotze et al. 2019; Tittensor et al. 2021). While considering a broader set of scenarios would be ideal, it was not possible in this instance as only the RCP8.5/SSP5-8.5 had been run by all models. Extending to another scenario was unfortunately not feasible due to computational intensity and capacity as FishMIP and ISIMIP are largely volunteer contributions unlike CMIP. The high emissions scenario samples a large range of global warming, and as many impacts scale approximately with global warming, impacts under a low scenario may be
192 193 194 195 196 197 198 199	tishing and did not run fishing simulations. We focussed on the RCP8.5/SSP5-8.5 high emissions scenario as the impacts of emission scenario have been previously explored for CMIP5 and CMIP6 (Lotze et al. 2019; Tittensor et al. 2021). While considering a broader set of scenarios would be ideal, it was not possible in this instance as only the RCP8.5/SSP5-8.5 had been run by all models. Extending to another scenario was unfortunately not feasible due to computational intensity and capacity as FishMIP and ISIMIP are largely volunteer contributions unlike CMIP. The high emissions scenario samples a large range of global warming, and as many impacts scale approximately with global warming, impacts under a low scenario may be similar to, just smaller than, a high scenario. For example, if 2.0 °C is reached earlier in RCP8.5
192 193 194 195 196 197 198 199 200	tishing and did not run fishing simulations. We focussed on the RCP8.5/SSP5-8.5 high emissions scenario as the impacts of emission scenario have been previously explored for CMIP5 and CMIP6 (Lotze et al. 2019; Tittensor et al. 2021). While considering a broader set of scenarios would be ideal, it was not possible in this instance as only the RCP8.5/SSP5-8.5 had been run by all models. Extending to another scenario was unfortunately not feasible due to computational intensity and capacity as FishMIP and ISIMIP are largely volunteer contributions unlike CMIP. The high emissions scenario samples a large range of global warming, and as many impacts scale approximately with global warming, impacts under a low scenario may be similar to, just smaller than, a high scenario. For example, if 2.0 °C is reached earlier in RCP8.5 than in RCP4.5, RCP8.5 might still give a fair estimate of the impacts of 2.0 °C, irrespective of

considered, as they can lead to different ecosystem consequences. FishMIP models simulated
climate change scenarios by incorporating relevant outputs from ESMs as MEM forcing
variables, such as temperature, primary productivity, phytoplankton biomass, zooplankton
biomass, acidity, oxygen concentration, and water velocities (Tables 1, 2; Figures 2, 3).

206

## 207 *Model comparisons*

208 We used regional model spatial domains to subset global model simulation output for 209 corresponding grid cells, such that the geographical domains were identical (https://github.com/Fish-MIP/Regional v Global). Following previous FishMIP studies (Lotze et 210 al. 2019; Tittensor et al. 2021), for each combination of CMIP, ESM, MEM, and region, we 211 212 calculated the percent change in total consumer biomass (all consumers of trophic level > 1, 213 vertebrates and invertebrates) for 2090-2099 relative to 1990-1999. To evaluate model 214 agreement among global and regional scale models in each region, we calculated the 215 proportion that the regional model time series projection fell within the range of the global model 216 ensemble, the root mean squared error (RMSE), and Spearman correlation. 217 218 Drivers of biomass change

To explore relationships between ESM environmental forcing variables and MEM total consumer biomass, for each region, model, and year combination, we calculated change in SST and NPP (relative to 1990-1999) to compare with the corresponding change in total consumer biomass (relative to 1990-1999). For each model, we calculated the amount of variation in delta total consumer biomass that was explained by delta NPP or delta SST.

224

225 3. Results

226 3.1 – Climate forcing variable projections

227 3.1.1 – Sea surface temperature

228 For both CMIP5 and CMIP6 simulation rounds, SST was projected to increase in all 229 regions (Figures 2, 3). For CMIP5, GFDL projections were warmer in three regions, while there was overlap in four regions (Figure 2). GFDL projections were higher in 2100 in three regions, 230 231 while IPSL projections were higher in four regions (Figure 2; Table 3). Interannual variability was 232 similar for GFDL and IPSL projections, with exceptions for greater variability in the GFDL projection for the Humboldt Current and IPSL projections for the Cook Strait and East Bass 233 234 Strait (Figure 2). The magnitude of SST increase from 1950 – 2100 was greater in IPSL projections in all seven regions (Figure 2). For CMIP6, GFDL projections were warmer in three 235 regions, an IPSL projection was warmer in one region, with overlap between a GFDL and IPSL 236 projection in one region (Figure 3). GFDL projections were higher in 2100 in three regions, while 237 238 IPSL projections were higher in two (Figure 3; Table 3). Interannual variability was similar for 239 GFDL and IPSL projections except in the East Bass Strait, where the GFDL projection showed 240 greater variability (Figure 3). The magnitude of SST increase from 1950 – 2100 was greater for 241 a GFDL projection in one region, IPSL projections in two regions, and similar in two regions 242 (Figure 3).

243

244 3.1.2. – Net primary productivity

For both CMIP5 and CMIP6 projections, there was variability in direction of NPP change 245 246 by 2100 (Figures 2, 3). For CMIP5, GFDL NPP projections were higher in six regions, with one region showing overlap (Figure 2). GFDL projections were higher in 2100 in all regions (Figure 247 2: Table 3). There were similar amounts of interannual variability for GFDL and IPSL 248 projections, except in the Baltic Sea and East Bass Strait, where IPSL projections were more 249 250 variable (Figure 2). IPSL projections indicated greater NPP declines from 1950-2100 in four of 251 seven regions, while in the other three regions there were no strong trends (Figure 2). For 252 CMIP6, GFDL projections of NPP were greater than IPSL projections in all five regions (Figure 253 3). There was similar interannual variability for GFDL and IPSL projections, except in the East

Bass Strait where the GFDL projection showed greater variability (Figure 3). The magnitude of
change in NPP was variable, with GFDL projected decreases and IPSL increases in Cook
Strait, East Bass Strait, and Hawaii regions (Figure 3). Both GFDL and IPSL projections
indicated increases in NPP for the Southern Benguela and Eastern Bering Sea, with greater
increase in the IPSL projection for the Southern Benguela and greater increase in the GFDL
projection for the Eastern Bering Sea (Figure 3).

260

## 261 3.2 – Global & regional model ensemble projections

262 On average, global models projected greater biomass declines than regional models. 263 For CMIP5 simulations, the average biomass decline at the end of the century was 6% for 264 regional models compared to 18% for global models (Figure 4, S2; Table 4). For CMIP6 265 simulations, on average regional models projected a decline of 18% at the end of the century, 266 while global models projected a decline of 27% (Figure 5, S3, Table 4).

267 Global model ensemble projections often produced a decline in biomass at the end of the century. For CMIP5, only 14% of the global ensemble projections showed increases (n = 268 269 2/14; biomass increases for GFDL simulations in East Bass Strait and Mediterranean Sea; Table 4; Figures 2, 4, 7). In contrast, regional models projected biomass increases at the end of 270 271 the century in 50% of CMIP5 simulations (n = 7/14, for 7 regions and 2 ESMs; Table 4; Figures 272 2, 4, 6). For CMIP6 simulations, global model ensemble projections always projected biomass declines at the end of the century (n=12), while regional models projected biomass increases in 273 274 33% of simulations (n = 4/12; IPSL simulations for Cook Strait and East Bass Strait; GFDL and 275 IPSL simulations for Southern Benguela EwE; Table 4; Figures 3, 5, 7). However, it should be 276 noted that some global models in some regions projected the opposite direction of change as 277 the global model ensemble (Figures 6, 7).

For some regions, biomass projections from regional models showed different trends than projections from the global ensemble (Figures 4, 5). This was observed in both GFDL and 280 IPSL simulations and in both CMIP5 and CMIP6 simulation rounds (Figures 4, 5). In the CMIP5 281 simulation round, disagreement in direction of biomass change was observed in the Cook Strait and Southeast Australia for both GFDL and IPSL simulations (Figure 4) and the East Bass Strait 282 283 for GFDL simulations (Figure 4; Table 4). In the CMIP6 simulation round, differences in 284 projected direction of change were observed in the Cook Strait and the East Bass Strait for IPSL simulations and the Southern Benguela EwE model for both GFDL and IPSL simulations (Figure 285 7; Table 4). In the Eastern Bering Sea and Hawaii, for both GFDL and IPSL simulations and in 286 287 the East Bass Strait for GFDL simulations, biomass trends were very different in terms of 288 magnitude of change and variability, with the regional models projecting greater biomass declines in the Eastern Bering Sea, while the opposite response was observed in Hawaii and 289 East Bass Strait (Figure 5). 290

291 Regional model biomass change projections were within the range of the global model 292 ensemble on average for 43% of the time series in CMIP5 and 36% of the time series in CMIP6 293 (Figures 4, 5; Table S1). In the CMIP5 simulation round, less than half of the regional model time series was within the global model ensemble range for: Baltic Sea – IPSL, Cook Strait – 294 295 GFDL and IPSL, East Bass Strait – IPSL, Humboldt Current – GFDL and IPSL, Mediterranean 296 Sea – GFDL, North Sea – GFDL, and SE Australia – both GFDL and IPSL (Figures 4, 6; Table S1). In the CMIP6 simulation round, regions where the regional model was within the range of 297 298 the global ensemble for less than half of the time series were: Cook Strait – IPSL, East Bass Strait – GFDL and IPSL; East Bering Sea – GFDL and IPSL; Hawaii – GFDL and IPSL; and 299 300 Southern Benguela Atlantis for GFDL and IPSL and Southern Benguela EwE for GFDL (Figures 5, 7; Table S1). On average, for CMIP5, this was 42.7% for GFDL and 44.1% for IPSL 301 for an overall average of 43.4% (Table S1). For CMIP6, this was 38% for GFDL, 34% for IPSL 302 303 for an overall average of 36% (Table S1).

304

305 3.3 – CMIP simulation round variability

For both global and regional models, greater average projected biomass changes at the end of the 21st century relative to 1990-1999 were observed in CMIP6 than in CMIP5 (23% vs. 12%, respectively; Figures 4, 5; Table 4). While the regions represented by both regional and global models differed in CMIP5 and CMIP6 simulation rounds, the two regions that participated in both rounds – Cook Strait and East Bass Strait – both showed greater average biomass declines in CMIP6 (Cook Strait: CMIP5 10% increase vs. CMIP6 28% decrease; East Bass Strait: CMIP5 3% decrease vs. CMIP6 12% decrease; Figures 4, 5; Table 4).

313

314 3.4 – ESM variability

On average, IPSL simulations produced greater biomass declines than GFDL 315 simulations (Figures 4-7, S4, S5; Table 4). In the CMIP5 simulation round, regional model GFDL 316 317 runs produced an average biomass increase of 10%, compared to a 22% decline for IPSL 318 simulations (Figures 4, 6, S4; Table 4). For global models in the CMIP5 simulation round, GFDL 319 runs produced an average biomass decline of 5% while IPSL runs produced an average 320 biomass decline of 31% (Table 4). For the CMIP6 simulation round, GFDL simulations produced 321 greater biomass declines than IPSL, although IPSL showed larger variability over time (Figures 322 5, 7, S5; Table 4). For regional models, GFDL simulations produced an average biomass decline of 26%, with 11% for IPSL (Table 4). For global models, GFDL simulations produced an 323 324 average biomass decline of 28% while IPSL simulations produced an average decline of 26% (Table 4). 325

Agreement between global and regional models was similar for GFDL and IPSL runs in the CMIP5 simulation round (average RMSE of 14.9 vs. 15.0, respectively; Table S2). For the CMIP6 simulation round, global and regional model agreement was better for GFDL simulations than IPSL (average RMSE of 18.3 vs. 24.5, respectively; Table S2).

330

331 3.5 – Temperature and net primary productivity relationships with total consumer biomass

332 All MEMs showed a negative relationship between change in SST and change in total consumer biomass, however the slopes of the linear regressions differed among models (Figure 333 8; Table S3). The greatest slopes were observed for mizer and Macroecological (-11.7 and -334 335 11.5, respectively) and the smallest slope was observed for Ecopath with Ecosim (EwE; -3.2). 336 The amount of variation in delta total consumer biomass that was explained by delta SST was greatest for DBPM ( $R^2 = 0.66$ ), while the smallest was observed for EwE ( $R^2 = 0.03$ ). Most 337 338 MEMs had a positive relationship between change in NPP and total consumer biomass, except 339 for Atlantis and mizer, both of which are regional models (Figure 9, Table S3). The model that had the greatest positive slope was EwE (2.3E+08), while the greatest negative slope was 340 observed for mizer (-4.3E+08; Figure 9, Table S3). The highest amount of variation in delta total 341 consumer biomass that was explained by delta NPP was observed for EwE ( $R^2 = 0.41$ ), while 342 the lowest value was observed for OSMOSE ( $R^2 = 0.04$ ; Figure 9). 343

344

345 Discussion

Our results indicate that climate change projections by global and regional marine ecosystem models covering the same location often differ in magnitude and sometimes direction of biomass change. On average, global models projected greater biomass declines than regional models, CMIP6 simulations projected greater biomass declines than CMIP5 simulations, and IPSL simulations projected greater biomass declines than GFDL simulations. Overall, regional model projections were within the range of global model ensembles for less than half of the time series.

Mismatches between global and regional models can be attributed to several factors. The first is how ecology and the multifaceted effects of climate drivers interact and are represented in each MEM. Global MEMs have been mostly developed for climate impact studies and tend to include more climate forcing variables compared with the regional models considered in this study (Table 1). For example, the divergent trends for Cook Strait and East 358 Bass Strait compared to global MEMs may reflect the fact that these two models were forced with primary production changes only, and do not include temperature effects on bioenergetics 359 (Table 2). The latter can lead to more marked declines in biomass (Carozza, 2019, Heneghan et 360 361 al. 2021) and we observed variable relationships between delta total consumer biomass and 362 delta SST and NPP (Figures 8,9). On the other hand, including more regional species-specific 363 detail in temperature responses (as in the Hawaii therMizer model), can dampen or lead to 364 antagonistic effects as they propagate from individual physiological to community levels. An 365 analysis of global FishMIP MEMs showed that incorporating temperature effects led to biomass 366 changes of -35% to 3% while simulating climate change through low trophic level effects (primary production; phytoplankton and/or zooplankton biomass or production) produced 367 biomass changes of -17% to 15% (Heneghan et al. 2021). 368

369 The ecological complexity of global and regional MEMs also differs, as regional models 370 often include more food web interactions and functional diversity than global models (Tables 1, 371 2). Using the same underlying regional size-based modelling framework and only temperature 372 altered, fish community responses were not generalizable among regions and were dependent 373 on the species or functional group position in the food web and food web structure (Reum et al. 374 2024). Compared to general trait-based food web model configurations, more detailed regional food web structures led to damped effects of warming (Reum et al. 2024). This could explain 375 376 why global FishMIP MEMs overestimated the observed impacts in an analysis that used the 377 2003 European heat wave as a case study and compared global FishMIP MEM hindcasts of 378 ocean biomass to observations (Schewe et al. 2019). In a comparison of global and regional 379 MEMs in SE Australia, it was not the climate drivers but the representation of the ecology that proved critical in explaining variation in model projections (Pethybridge et al. 2020; Fulton 380 381 2021). The regional Atlantis modelling framework, as applied in SE Australia, includes all the 382 same climate drivers as the global models but produced divergent results with global MEMs. 383 Differences between the Atlantis projections and those from global MEMs had many causes -

384 from variable representation of species or spatial sub-domains of the model (Pethybridge et al. 2020; Fulton 2021). However, one common dynamic leading to divergence between this 385 regional MEM and the overlapping global MEMs was due to ecologically mediated interactions. 386 387 The trophic resolution of Atlantis included more detail and more feedback pathways than in the 388 global MEMs. This meant that situations arose where decreased predation or competition acting 389 on a structurally important species for the food web outweighed direct climate effects on that 390 species (Pethybridge et al. 2020; Fulton 2021). In other instances, movement and 391 ontogenetically mediated processes and connectivity were important (Pethybridge et al. 2020; 392 Fulton 2021).

It has also been noted that among three commonly used regional MEM platforms -393 Atlantis, EwE, and OSMOSE - there is system specificity in the degree of convergence and 394 395 divergence in projections (Smith et al. 2011). For example, running the same maximum 396 sustainable yield (MSY) fisheries simulations in each of these modelling frameworks in four 397 regional ecosystems (Benguela, Humboldt, and California Current systems and Southeast 398 Australia) showed clear variability in projected ecosystem responses (Smith et al. 2011). 399 However, in other instances various combinations of Atlantis, OSMOSE and Ecopath with 400 Ecosim models have projected similar general patterns of change to fishing (Travers et al., 2010; Forrest et al. 2015; Smith et al. 2015; Ortega-Cisneros et al. 2018), where aggregated 401 402 properties showed more consistency across models than species level variables (Ortega et al. 403 2018). The specific degree of model responsiveness to perturbation is also typically system 404 specific, though in general terms the Atlantis modelling framework is less sensitive to ecosystem perturbations, especially environmentally driven events, than Ecopath with Ecosim and 405 OSMOSE (Fulton and Smith 2004; Smith et al. 2011). Model sensitivity is an important 406 407 consideration when using models to guide policy advice, such as that provided by the IPCC at the global scale, but also for interpreting models and providing strategic fisheries management 408

guidance as provided by the FAO at national, regional, and local scales (Blanchard andNovaglio 2024).

An additional factor contributing to global and regional model mismatches is the coarse 411 412 resolution of coastal regions in global ESMs and MEMs. Global models often poorly represent waters <50 m depth, and at the 1 ° grid size scale (~100 km by 100 km at the equator) fail to 413 414 capture fine-scale coastal and shelf processes such as eddies and upwelling - important for 415 nutrient supply, and production of phytoplankton, zooplankton, higher trophic levels, and 416 fisheries (Laurent et al. 2021; Pozo-Buil et al. 2021). Two approaches to increase resolution 417 are: statistical downscaling to a higher resolution grid – which will be influenced by the ESM that it was downscaled from (Lange 2019; Oliveros-Ramos et al. in revision); or by dynamical 418 419 downscaling with a regional biogeochemical model or a regional ocean modelling system 420 (ROMS; Laurent et al. 2021; Pozo-Buil et al. 2021). Regional MEMs have faced challenges 421 incorporating highly resolved spatial data as drivers of change. Some regional MEMs have been 422 developed without explicit spatial resolution; instead using implicit representation to structure 423 food webs by depth or other influencing process. There is also a tradeoff between downscaled 424 models having higher spatial resolution but poorer estimates of uncertainty because their 425 boundary conditions are often driven by only one ESM, and commonly few emission scenarios 426 are included (Pozo-Buil et al. 2021). Regional models can also be limited by an inaccurate 427 representation of boundary conditions, such as the import and export of water and biomass to 428 the model domain (across both land/riverine and oceanic boundaries).

Comparisons of CMIP5 ESM projections to regional observations of environmental variables have concluded that coarsely resolved ESMs failed to accurately capture complex patterns of circulation and elemental fluxes on the shelves along ocean margins of the northwest Atlantic shelf (Laurent et al. 2021). ESMs underestimated observed chlorophyll and nitrate, while a regional ocean modelling system (ROMS) biogeochemical model with higher spatial resolution reproduced observed trends better (Laurent et al. 2021). Similar mismatches 435 between model projections and observed values were observed in regions we investigated, 436 such as for the Humboldt Current, where ESM projections of NPP were half the magnitude compared to observed values – which was bias-corrected through statistical downscaling 437 438 (Lange et al. 2019; Oliveros-Ramos et al. 2024). Global and regional model mismatches have 439 also been observed in comparisons of hydrological models, where global models not calibrated 440 to regional observations failed to reproduce regional trends, with the recommendation that 441 regional models were more appropriate for regional water management (Gosling et al. 2016; 442 Hattermann et al. 2017).

443 Model calibration is a key source of variation among global and regional models. Fitting models to time series of fisheries and biomass surveys has been a focus of many regional 444 models since their inception (Bentley et al., 2024) but is only just beginning for global models. 445 446 Regional models often use fisheries catch and fisheries independent survey data in the region 447 (Maureaud et al. 2021; Maureaud et al. 2024). Global models are often limited by observational data for calibration, as only exclusive economic zone (EEZ) or large marine ecosystem (LME) 448 scale catch data are available. Data at these scales have many uncertainties, including how 449 450 fishing effort is spatially allocated and representativeness of total system removals (Watson 451 2017; Rousseau et al. 2019; Rousseau et al. 2024).

452 As ESM accuracy improves and computing power increases, the spatial resolution of 453 ESMs will increase and processes within coastal and shelf regions will be better resolved and more consistent with regional oceanographic models. ESM runs undertaken at a 0.25 ° grid 454 455 scale do a better job of representing coastal and shelf features, such as fine scale eddies, currents, and upwelling. Such changes in resolution can vastly improve the representation of 456 ecologically relevant features, such as production hotspots, both in global but also regional 457 458 modelling initiatives (Matear et al. 2013). As global scale ESMs and MEMs at both global and 459 regional scales continue to evolve, a central goal of FishMIP is to understand sources of variation to build confidence in projections of climate change impacts on marine ecosystems at 460

regional scales and to provide guidance about which scales different models can be applied for
adaptation and mitigation planning. We note that the test undertaken here, comparing
projections at the end of the 21st century looks at the end point not the transition pathways, and
that we may be missing divergence that has real meaning at temporal scales meaningful for
decision makers.

A key source of uncertainty that has not yet been explored with global and regional 466 467 MEMs within FishMIP is socioeconomics. Market, management, and policy responses could be 468 more important than climate change for the future of fish populations (Cheung et al. 2021, 2024). To date, FishMIP has focussed on the climate change impact and treated the 469 socioeconomic impact simply by holding fishing levels constant at 2005 or 2015 levels or with a 470 471 no-fishing scenario. The SSPs were not specifically developed for marine systems and are not 472 directly applicable to socioeconomic factors for ocean systems. To address this important 473 consideration, FishMIP has established a Scenarios Working Group to develop ocean system 474 pathways (OSPs) of future fishing effort and ocean usage, based on the SSP scenarios, with 475 the intention that they be used in combination with RCP emissions scenarios in future FishMIP 476 simulation rounds (Maury et al. in revision).

477 The value of model ensembles lies in building confidence in model projections through 478 the exploration of multiple models. When the exact processes and structures to represent 479 observations are unknown, comparison of varying model formulations provides a hypothesis 480 testing approach. If models agree in projections, there is greater confidence that key processes 481 and structures have been represented. Conversely, large variation in model projections points to missing knowledge. The global and regional models explored here employ different approaches 482 to represent marine ecosystems due to imperfect knowledge. Our study highlights areas for 483 484 future exploration but cautions that there is high uncertainty about how ecology and eco-485 evolutionary processes will unfold in rapidly changing marine environments. Of high importance are additional physical processes that function at scales not represented by the resolution of the 486

487 ESMs but are implicit in regional models that have been bias corrected. Our comparison of 488 global and regional MEMs highlights that there are key uncertainties for climate change projections of biomass change at regional scales. While management and conservation 489 490 organizations require this information for adaptation and mitigation planning, it should be 491 recognized that projections at these scales remain uncertain in many regions. FishMIP has 492 plans to continue to address these uncertainties at regional scales to build confidence in 493 projections of climate change impacts on marine ecosystems worldwide (Ortega-Cisneros et al. 494 In revision; Murphy et al. In revision).

- 495
- 496

497 Acknowledgements

498 We thank CM Petrik for her FishMIP coordination and for running the FEISTY simulations. TDE, 499 HKL, DT acknowledge funding support from the Natural Sciences and Engineering Research 500 Council of Canada (NSERC) and the Inter-Sectoral Impact Model Intercomparison Project 501 (ISIMIP). MC and JS acknowledge funding from the Spanish National Project ProOceans 502 (RETOS-PID2020-118097RB-I00) and MC acknowledges support by the "Severo Ochoa Centre 503 of Excellence" accreditation (CEX2019-000928-S) to the Institute of Marine Science (ICM-504 CSIC). KOC acknowledges support from the One Ocean Hub and the National Research 505 Foundation of South Africa (Grant 136481). KOC also acknowledges the resources from the Cluster for High Performance Computing-CSIR. LJS acknowledges funding and support from 506 the EU-Horizon 2020 project "TRIATLAS" (Grant No. 817578) and the One Ocean Hub. ROR 507 thanks the Pôle de Calcul et de Données Marines (PCDM) for providing access to their 508 509 DATARMOR computing resources (https://wwz.ifremer.fr/pcdm/Equipement). YJS 510 acknowledges funding support from the Biodiversa and Belmont Forum project SOMBEE 511 (BiodivScen ERA-Net COFUND programme, ANR contract n°ANR-18-EBI4-0003-01) and the

512 Pew Marine Fellows programme.

- 514 Open Research
- 515 The R scripts used to execute the analyses in the paper can be found at https://github.com/Fish-
- 516 MIP/Regional v Global. The simulations used for the study are available at ISIMIP data
- 517 repository (https://data.isimip.org/) under the tree 'ISIMIP2a/OutputData/marine-
- 518 fishery\_regional' and 'ISIMIP3a/OutputData/marine-fishery\_regional', for instructions to get
- 519 access see https://www.isimip.org/gettingstarted/data-access/.
- 520
- 521 References
- 522 Barange, M., Merino, G., Blanchard, J., Scholtens, J., Harle, J., Allison, E. H., Allen, J. I., Holt,
- 523 J., & Jennings, S. (2014). Impacts of climate change on marine ecosystem production in
- societies dependent on fisheries. Nature Climate Change, 4, 211–216.
- 525 https://doi.org/10.1038/nclimate2119
- 526
- 527 Bentley, J. W., Coll, M., Chagaris, D., Christensen, V., Heymans, J. J., Serpetti, N., & Walters,
- 528 C. J. (2024). Calibrating ecosystem models to support marine ecosystem-based management.
- 529 ICES Journal of Marine Science.
- 530
- 531 Blanchard, J. L., & Novaglio, C., eds. (2024). Climate change risks to marine ecosystems and
- 532 fisheries Projections to 2100 from the Fisheries and Marine Ecosystem Model
- 533 Intercomparison Project. FAO Fisheries and Aquaculture Technical Paper, No. 707, Rome,
- 534 FAO. https://doi.org/10.4060/cd1379en

- Blanchard, J. L., Jennings, S., Holmes, R., Harle, J., Merino, G., Allen, J. I., Holt, J., Dulvy, N.
- 537 K., & Barange, M. (2012). Potential consequences of climate change for primary production and

- 538 fish production in large marine ecosystems. Philosophical Transactions of the Royal Society B:
- 539 Biological Sciences, 367, 2979–2989. https://doi.org/10.1098/rstb.2012.0231
- 540
- 541 Blanchard, J. L., Watson, R. A., Fulton, E. A., Cottrell, R. S., Jennings, S. (2017). Linked
- sustainability challenges and trade-offs among fisheries, aquaculture, and agriculture. Nature
  Ecology & Evolution, 1, 1240–1249.
- 544
- 545 Blanchard, J. L., Novaglio, C., Maury, O., Harrison, C. S., Petrik, C. M., Fierro-Arcos, L. D.,
- 546 Ortega-Cisneros, K., Bryndum-Buchholz, A., Eddy, T. D., Heneghan, R., Roberts, K., Schewe,
- J., Bianchi, D., Guiet, J., van Denderen, P. D., Palacios-Abrantes, P., Liu, X., Stock, C.,
- 548 Rousseau, Y., Büchner, M., Adekoya, E. O., Bulman, C., Cheung, W. W. L., Christensen, V.,
- 549 Coll, M., Capitani, L., Datta, S., Fulton, E. A., Fuster, A., Garza, V., Lengaigne, M., Lindmark,
- 550 M., Murphy, K., Ouled-Cheikh Bonan, J., Prasad, S. S., Oliveros-Ramos, R., Reum, J. C.,
- 551 Rynne, N., Scherrer, K. J. N., Shin, Y. J., Steenbeek, J., Woodworth-Jefcoats, P., Wu, Y. L., &
- 552 Tittensor, D. P. (2024). Detecting, attributing, and projecting global marine ecosystem and
- 553 fisheries change: FishMIP 2.0. AGU preprint,
- 554 https://doi.org/10.22541/essoar.170594183.33534487/v1
- 555
- Bopp, L., Resplandy, L., Orr, J. C., Fichefet, T., Hov, Ø., Eyring, V., Houghton, R. A., & Madec,
  G. (2013). Multiple stressors of ocean ecosystems in the 21st century: projections with CMIP5
  models. Biogeosciences, 10, 6225–6245.
- 559
- 560 Boyce, D. G., Lotze, H. K., Tittensor, D. P., Carozza, D. A., & Worm, B. (2020). Future ocean
- biomass losses may widen socioeconomic equity gaps. Nature Communications, 11, 2235.
- 562

563	Bryndum-Buchholz, A., Tittensor, D. P., Blanchard, J. L., Cheung, W. W. L., Coll, M., Galbraith,
564	E. D., Jennings, S., Maury, O., & Lotze, H. K. (2019). Twenty-first-century climate change
565	impacts on marine animal biomass and ecosystem structure across ocean basins. Global
566	Change Biology. https://doi.org/10.1111/gcb.14512
567	
568	Carozza, D. A., Bianchi, D., & Galbraith, E. D. (2016). The ecological module of BOATS-1.0: A
569	bioenergetically constrained model of marine upper trophic levels suitable for studies of fisheries
570	and ocean biogeochemistry. Geoscientific Model Development, 9, 1545–1565.
571	https://doi.org/10.5194/gmd-9-1545-2016
572	
573	Carozza, D. A., Bianchi, D., & Galbraith, E. D. (2019). Metabolic impacts of climate change on
574	marine ecosystems: Implications for fish communities and fisheries. Global Ecology and
575	Biogeography, 28(2), 158-169.
576	
577	Cheung, W. W. L., Watson, R., & Pauly, D. (2009). Projecting global marine biodiversity impacts
578	under climate change scenarios. Fish and Fisheries, 10, 235–251.
579	
580	Cheung, W. W. L., Lam, V. W. Y., Sarmiento, J. L., Kearney, K., Watson, R., & Pauly, D. (2010).
581	Large-scale redistribution of maximum fisheries catch potential in the global ocean under
582	climate change. Global Change Biology, 16, 24–35.
583	
584	Cheung, W. W. L., Reygondeau, G., Frölicher, T. L., Lam, V. W. Y., Oyinlola, M. A., & Sumaila,
585	U. R. (2016). Structural uncertainty in projecting global fisheries catches under climate change.
586	Ecological Modelling, 325, 57–66.
587	

588	Cheung, W. W. L., Frölicher, T. L., Lam, V. W., Oyinlola, M. A., Reygondeau, G., Sumaila, U.
589	R., Tai, T. C., Teh, L. C., & Wabnitz, C. C. (2021). Marine high-temperature extremes amplify
590	the impacts of climate change on fish and fisheries. Science Advances, 7(40), eabh0895.
591	
592	Cheung, W. W. L., Palacios-Abrantes, J., & Roberts, S. M. (2024). Projecting contributions of
593	marine protected areas to rebuild fish stocks under climate change. npj Ocean Sustainability,
594	3(1), 11.
595	
596	Christensen, V., & Walters, C. J. (2004). Ecopath with Ecosim: methods, capabilities and
597	limitations. Ecological Modelling, 172, 109–139. https://doi.org/10.1016/j.ecolmodel.2003.09.003
598	
599	Christensen, V., Coll, M., Steenbeek, J., Buszowski, J., Chagaris, D., & Walters, C. J. (2014).
600	Representing variable habitat quality in a spatial food web model. Ecosystems, 17(8), 1397-
601	1412.
602	
603	Christensen, V., Coll, M., Buszowski, J., Cheung, W. W. L., Frölicher, T., Steenbeek, J., Stock,
604	C. A., Watson, R. A., & Walters, C. J. (2015). The global ocean is an ecosystem: simulating
605	marine life and fisheries. Global Ecology and Biogeography, 24, 507–517.
606	https://doi.org/10.1111/geb.12281
607	

- 608 Cinner, J. E., Caldwell, I., Thiault, L., Ben, J., Blanchard, J. L., Coll, M., Diedrich, A., Eddy, T.
- D., Everett, J. D., Folberth, C., Gascuel, D., Guiet, J., Gurney, G. G., Heneghan, R. F.,
- Jägermeyr, J., Jiddawi, N., Lahari, R., Kuange, J., Liu, W., Maury, O., Müller, C., Novaglio, C.,
- Palacios-Abrantes, J., Petrik, C. M., Rabearisoa, A., Tittensor, D. P., Wamukota, A., & Pollnac,
- R. (2022). The potential impacts of climate change on agriculture and fisheries production in 72

- tropical coastal communities. Nature Communications, 13, 30991.
- 614 https://doi.org/10.1038/s41467-022-30991-4
- 615
- 616 Coll, M., Piroddi, C., Kaschner, K., Ben Rais Lasram, F., Steenbeek, J., Aguzzi, J., Ballesteros,
- E., Bianchi, C. N., Corbera, J., Dailianis, T., Danovaro, R., Estrada, M., Froglia, C., Galil, B. S.,
- Gasol, J. M., Gertwagen, R., Gil, J., Guilhaumon, F., Kesner-Reyes, K., Kitsos, M.-S.,
- 619 Koukouras, A., Lampadariou, N., Laxamana, E., López-Fé de la Cuadra, C. M., Lotze, H. K.,
- Martin, D., Mouillot, D., Oro, D., Raicevich, S., Rius-Barile, J., Saiz-Salinas, J. I., San Vicente,
- 621 C., Somot, S., Templado, J., Turon, X., Vafidis, D., Villanueva, R., & Voultsiadou, E. (2010). The
- biodiversity of the Mediterranean Sea: estimates, patterns and threats. PLoS ONE, 5(8),
- 623 e11842. https://doi.org/10.1371/journal.pone.0011842
- 624
- 625 Coll, M., Steenbeek, J., Pennino, G. M., Buszowski, J., Kaschner, K., Lotze, H. K., Rousseau,
- Y., Tittensor, D., Walters, C., Watson, R. A., & Christensen, V. (2020). Advancing global
- 627 ecological modelling capabilities to simulate future trajectories of change in marine ecosystems.
- 628 Frontiers in Marine Science, 7, 567877.
- 629
- 630 Coll, M., Bellido, J. M., Pennino, M. G., Albo-Puigserver, M., Báez, J. C., Christensen, V.,
- 631 Corrales, X., Fernández-Corredor, E., Giménez, J., Julià, L., Lloret-Lloret, E., Macias, D., Ouled-
- 632 Cheikh, J., Ramírez, F., Sbragaglia, V., & Steenbeek, J. (2024). Retrospective analysis of the
- 633 pelagic ecosystem of the Western Mediterranean Sea: drivers, changes and effects. Science of
- 634 The Total Environment, 167790.
- 635
- du Pontavice, H., Gascuel, D., Reygondeau, G., Stock, C., & Cheung, W. W. L. (2021). Climate-
- 637 induced decrease in biomass flow in marine food webs may severely affect predators and

- 638 ecosystem production. Global Change Biology, 27, 2608–2622.
- 639 https://doi.org/10.1111/gcb.15576
- 640
- 641 Eddy, T. D., Pitcher, T. J., MacDiarmid, A. B., Byfield, T. T., Jones, T., Tam, J., Bell, J. J., &
- Gardner, J. P. A. (2014). Lobsters as keystone: Only in unfished ecosystems? Ecological
  Modelling, 275, 48-72.
- 644
- Eddy, T. D. (2019). Building confidence in projections of future ocean capacity. In Predicting
  Future Oceans (pp. 69-76). Elsevier.
- 647
- Eddy, T. D., Bopp, L., Bulman, C. M., Cheung, W. W. L., Coll, M., Fu, C., Fulton, E. A.,
- Galbraith, E., Lotze, H. K., Lynam, C., Mackinson, S., Niiranen, S., Oliveros-Ramos, R., Piroddi,
- 650 C., Steenbeek, J., Tittensor, D. T., Walker, N. D., Schewe, J., Volkholz, J., & Büchner, M.
- 651 (2023). ISIMIP2b Simulation Data from the Regional Fisheries & Marine Ecosystems Sector
- 652 (v2.0). ISIMIP Repository. https://doi.org/10.48364/ISIMIP.767747
- 653
- Eddy, T. D., Ortega-Cisneros, K., Bulman, C., Reum, J., Shannon, L., Woodworth-Jefcoats, P.,
- Harrison, C. S., Maury, O., Petrik, C., Novaglio, C., Heneghan, R. F., Bryndum-Buchholz, A.,
- Tittensor, D. P., & Blanchard, J. L. (2024). ISIMIP3b Simulation Data from the Regional
- 657 Fisheries & Marine Ecosystems Sector. https://doi.org/10.48364/ISIMIP.128964
- 658
- 659 Eyring, V., Bony, S., Meehl, G. A., Senior, C. A., Stevens, B., Stouffer, R. J., & Taylor, K. E.
- 660 (2016). Overview of the Coupled Model Intercomparison Project Phase 6 (CMIP6) experimental
- design and organization. Geoscientific Model Development, 9(5), 1937-1958.
- 662

FAO. (2022). The State of World Fisheries and Aquaculture 2022. Towards BlueTransformation. Rome: FAO.

665

Forrest, R. E., Savina, M., Fulton, E. A., & Pitcher, T. J. (2015). Do marine ecosystem models
give consistent policy evaluations? A comparison of Atlantis and Ecosim. Fisheries Research,
167, 293-312.

669

Free, C. M., Thorson, J. T., Pinsky, M. L., Oken, K. L., Wiedenmann, J., & Jensen, O. P. (2019).

671 Impacts of historical warming on marine fisheries production. Science, 363(6430), 979-983.

672

- 673 Frieler, K., Volkholz, J., Lange, S., Schewe, J., Mengel, M., Rivas López, M. del R., Otto, C.,
- Reyer, C. P. O., Karger, N., Malle, J. T., Treu, S., Menz, C., Blanchard, J. L., Harrison, C. S.,
- Petrik, C. M., Eddy, T. D., Ortega-Cisneros, K., Novaglio, C., Rousseau, Y., Watson, R. A.,
- Stock, C., Liu, X., Heneghan, R., Tittensor, D., Maury, O., Büchner, M., Vogt, T., Wang, T., Sun,
- F., Sauer, I. J., Koch, J., Vanderkelen, I., Jägermeyr, J., Müller, C., Klar, J., Vega del Valle, I. D.,
- Lasslop, G., Chadburn, S., Burke, E., Gallego-Sala, A., Smith, N., Chang, J., Hantson, S.,
- Burton, C., Gädeke, A., Li, F., Gosling, S., Müller Schmied, H., Hattermann, F., Wang, J., Yao,
- 680 F., Hickler, T., Marcé, R., Pierson, D., Thiery, W., Mercado-Bettín, D., Forrest, M., & Bechtold,
- 681 M. (2024). Scenario set-up and forcing data for impact model evaluation and impact attribution
- 682 within the third round of the Inter-Sectoral Model Intercomparison Project (ISIMIP3a).
- 683 Geoscientific Model Development, 17, 1–51.

684

- Fulton, E. A., & Smith, A. D. M. (2004). Lessons learnt from a comparison of three ecosystem
- models for Port Phillip Bay, Australia. African Journal of Marine Science, 26, 219–243.

- Fulton, E. A., Link, J. S., Kaplan, I. C., Savina-Rolland, M., Johnson, P., Ainsworth, C., Horne,
- P., Gorton, R., Gamble, R. J., Smith, A. D. M., & Smith, D. C. (2011). Lessons in modelling and
- 690 management of marine ecosystems: the Atlantis experience. Fish and Fisheries, 12, 171–188.
- 691 https://doi.org/10.1111/j.1467-2979.2011.00412.x
- 692
- Fulton, E. A., Hobday, A. J., Pethybridge, H., Blanchard, J., Bulman, C., Butler, I., Cheung, W.,
- 694 Gorton, B., Hutton, T., Lozano-Montes, H., Matear, R., Pecl, G., Villanueva, C., & Zhang, X.
- 695 (2018). Decadal scale projection of changes in Australian fisheries stocks under climate change.
- 696 CSIRO Report to FRDC, Project No: 2016/139.
- 697
- 698 Gascuel, D. (2005). The trophic-level based model: A theoretical approach of fishing effects on
- marine ecosystems. Ecological Modelling, 189, 315–332.
- 700 https://doi.org/10.1016/j.ecolmodel.2005.03.019
- 701
- Gascuel, D., Guénette, S., & Pauly, D. (2011). The trophic-level based ecosystem modelling
- approach: Theoretical overview and practical uses. ICES Journal of Marine Sciences, 68, 1403-
- 704 1416. https://doi.org/10.1093/icesjms/fsr062
- 705
- Gosling, S. N., Zaherpour, J. J., Mount, N. J., Hattermann, F. F., Dankers, R., Arheimer, B., ...
- 707 Zhang, X. (2016). A comparison of changes in river runoff from multiple global and catchment-
- scale hydrological models under global warming scenarios of 1°C, 2°C and 3°C. Climatic
- 709 Change, 1–19. https://doi.org/10.1007/s10584-016-1773-3
- 710
- 711 Guibourd de Luzinais, V., du Pontavice, H., Reygondeau, G., Barrier, N., Blanchard, J. L.,
- Bornarel, V., Büchner, M., Cheung, W. W. L., Eddy, T. D., Everett, J. D., Guiet, J., Harrison, C.
- S., Maury, O., Novaglio, C., Petrik, C. M., Steenbeek, J., Tittensor, D. P., & Gascuel, D. (2023).

Trophic amplification: A model intercomparison of climate-driven changes in marine food webs.
PLOS ONE, 18, e0287570.

716

- Hattermann, F. F., Gosling, S. N., Dankers, R., & Knox, J. C. (2017). Cross-scale
- intercomparison of climate change impacts simulated by global and regional hydrological
- models in eleven large river basins. Climatic Change, 141, 561–576.

720

- Heneghan, R. F., Everett, J. D., Sykes, P., Batten, S. D., Edwards, M., Takahashi, K., Suthers,
- I. M., Blanchard, J. L., & Richardson, A. J. (2020). A functional size-spectrum model of the
- global marine ecosystem that resolves zooplankton composition. Ecological Modelling, 435,
- 724 109265.

- Heneghan, R. F., Galbraith, E. D., Blanchard, J. L., Harrison, C. S., Barrier, N., Bulman, C.,
- 727 Cheung, W. W. L., Coll, M., Eddy, T. D., Erauskin-Extramiana, M., Everett, J. D., Fernandes-
- Salvador, J. A., Gascuel, D., Guiet, J., Maury, O., Palacios-Abrantes, J., Petrik, C. M., du
- Pontavice, H., Richardson, A. J., Steenbeek, J., Tai, T. C., Volkholz, J., Woodworth-Jefcoats, P.
- A., & Tittensor, D. P. (2021). Disentangling diverse responses to climate change among global
- marine ecosystem models. Progress in Oceanography, 193, 102560.
- 732 https://doi.org/10.1016/j.pocean.2021.102560
- 733
- 734 IPBES. (2019). Global assessment report on biodiversity and ecosystem services of the
- 735 Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services. Edited by
- E. S. Brondizio, J. Settele, S. Díaz, & H. T. Ngo. IPBES secretariat, Bonn, Germany.
- 737 https://doi.org/10.5281/zenodo.3831673
- 738

- 739 IPCC. (2023). Climate Change 2023: Synthesis Report. Contribution of Working Groups I, II,
- and III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change

[Core Writing Team, H. Lee, & J. Romero (eds.)]. IPCC, Geneva, Switzerland.

- 742 https://doi.org/10.59327/IPCC/AR6-9789291691647
- 743
- Jennings, S., & Collingridge, K. (2015). Predicting consumer biomass, size-structure,
- 745 production, catch potential, responses to fishing, and associated uncertainties in the world's
- marine ecosystems. PLoS One, 10, e0133794. https://doi.org/10.1371/journal.pone.0133794
- 747
- Kwiatkowski, L., Torres, O., Bopp, L., Aumont, O., Chamberlain, M., Christian, J. R., Dunne, J.
- P., Gehlen, M., Ilyina, T., John, J. G., Lenton, A., Li, H., Lovenduski, N. S., Orr, J. C., Palmieri,
- J., Santana-Falcón, Y., Schwinger, J., Séférian, R., Stock, C. A., ... Ziehn, T. (2020). Twenty-
- first century ocean warming, acidification, deoxygenation, and upper-ocean nutrient and primary
- production decline from CMIP6 model projections. Biogeosciences, 17(13), 3439–3470.
- 753 https://doi.org/10.5194/bg-17-3439-2020
- 754
- Lange, S. (2019). Trend-preserving bias adjustment and statistical downscaling with
  ISIMIP3BASD (v1.0). Geoscientific Model Development, 12, 3055–3070.
- 757
- Laurent, A., Fennel, K., & Kuhn, A. (2021). An observation-based evaluation and ranking of
- historical Earth System Model simulations for regional downscaling in the northwest North
- 760 Atlantic Ocean. Biogeosciences Discussions, 1–36. https://doi.org/10.5194/bg-2020-265
- 761
- Lotze, H. K., Tittensor, D. P., Bryndum-Buchholz, A., Eddy, T. D., Cheung, W. W. L., Galbraith,
- E. D., Barrier, N., Bianchi, D., Blanchard, J. L., Bopp, L., Büchner, M., Carozza, D. A.,
- Christensen, V., Coll, M., Dunne, J. P., Fulton, E. A., Jennings, S., Jones, M. C., Lehodey, P.,

765	Link, J. S., Mackinson, S., Maury, O., Niiranen, S., de Oliveira, J. A., Petrik, C. M., Steenbeek,
766	J., Stock, C. A., Verley, P., Volkholz, J., Watson, R. A., & Worm, B. (2019). Global ensemble
767	projections reveal trophic amplification of ocean biomass declines with climate change.
768	Proceedings of the National Academy of Sciences, 116(26), 12907–12912.
769	
770	Mackinson, S., & Daskalov, G. (2007). An ecosystem model of the North Sea to support an
771	ecosystem approach to fisheries management: description and parameterisation. Science
772	Series Technical Report, 142, Cefas Lowestoft, 196 pp.
773	
774	Mackinson, S., Platts, M., Garcia, C., & Lynam, C. (2018). Evaluating the fishery and ecological
775	consequences of the proposed North Sea multi-annual plan. PLoS ONE, 13, e0190015.
776	
777	Matear, R. J., Brown, J. N., Pecl, G. T., & Hobday, A. J. (2013). Oceanographic processes
778	affecting marine life. Journal of Geophysical Research: Oceans, 118(6), 2961–2976.
779	https://doi.org/10.1002/jgrc.20202
780	
781	Maureaud, A., Frelat, R., Pécuchet, L., Shackell, N., Mérigot, B., Pinsky, M. L., Amador, K., et
782	al. (2021). Are we ready to track climate-driven shifts in marine species across international
783	boundaries? A global survey of scientific bottom trawl data. Global Change Biology, 27(2), 220-
784	236. https://doi.org/10.1111/gcb.15404
785	
786	Maureaud, A. A., Palacios-Abrantes, J., Kitchel, Z., Mannocci, L., Pinsky, M. L., Fredston, A.,
787	Beukhof, E., et al. (2024). FISHGLOB_data: an integrated dataset of fish biodiversity sampled
788	with scientific bottom-trawl surveys. Scientific Data, 11, 24. https://doi.org/10.1038/s41597-023-
789	02866-w
790	

- 791 Maury, O. (2010). An overview of APECOSM, a spatialized mass balanced "Apex Predators
- 792 ECOSystem Model" to study physiologically structured tuna population dynamics in their
- rogress in Oceanography, 84(1-2), 113–117.
- 794 https://doi.org/10.1016/j.pocean.2009.09.013
- 795
- Maury, O., Tittensor, D. P., Eddy, T. D., Allison, E. H., Bari, T., Barrier, N., Campling, L.,
- 797 Cheung, W. W. L., Frieler, K., Fulton, E. A., Guillotreau, P., Heneghan, R. F., Lam, V. W. Y.,
- Leclère, D., Lengaigne, M., Lotze-Campen, H., Novaglio, C., Ortega-Cisneros, K., Schewe, J.,
- Shin, Y.-J., Sloterdijk, H., Squires, D., Sumaila, U. R., Tidd, A., van Ruijven, B. J., & Blanchard,
- J. L. (2024). The Ocean System Pathways (OSPs): A new scenario and simulation framework to
- 801 investigate the future of the world fisheries. ESS Open Archive,
- 802 https://doi.org/10.22541/essoar.171587166.60970779/v1
- 803
- Metcalfe, K., Vaz, S., Engelhard, G. H., Villanueva, M. C., & Coelho, R. (2015). Evaluating
- 805 conservation and fisheries management strategies by linking spatial prioritization software and
- ecosystem and fisheries modelling tools. Journal of Applied Ecology, 52(3), 665–674.
- 807
- Murphy, K., Fierro-Arcos, L. D., Rohr, T. W., Green, D. B., Novaglio, C., Baker, K., Ortega-
- Cisneros, K., Eddy, T. D., Harrison, C. S., Hill, S., Keith, P., Cataldo-Mendez, C., Petrik, C. M.,
- Pinkerton, M., Spence, P., Stollberg, I., Subramaniam, R., Trebilco, R., Tulloch, V., Palacios-
- Abrantes, J., Bestley, S., Bianchi, D., Boyd, P. W., Buchanan, P. J., Bryndum-Buchholz, A.,
- Coll, M., Corney, S. P., Datta, S., Everett, J. D., Forestier, R., Fulton, E. A., Galton-Fenzi, B. K.,
- B13 Guibourd de Luzinais, V., Heneghan, R., Mason, J. G., Maury, O., McMahon, C. R., Murphy, E.
- J., Richardson, A., Tittensor, D. P., Spillias, S., Steenbeek, J. G., Veytia, D., & Blanchard, J. L.
- 815 (2024). Developing a Southern Ocean marine ecosystem model ensemble to assess climate

- 816 risks and uncertainties. ESS Open Archive,
- 817 https://doi.org/10.22541/essoar.171580194.49771608/v1
- 818
- Niiranen, S., Yletyinen, J., Tomczak, M. T., Blenckner, T., Hjerne, O., MacKenzie, B. R., Müller-
- Karulis, B., Neumann, T., & Meier, H. E. M. (2013). Combined effects of global climate change
- and regional ecosystem drivers on an exploited marine food web. Global Change Biology,
- 822 19(11), 3327–3342. https://doi.org/10.1111/gcb.12309
- 823
- Novaglio, C., Bryndum-Buchholz, A., Tittensor, D. P., Eddy, T. D., Lotze, H. K., Harrison, C. S.,
- Heneghan, R., Maury, O., Ortega-Cisneros, K., Petrik, C. M., Roberts, K. E., & Blanchard, J. L.
- 826 (2024). The past and future of the fisheries and marine ecosystem model intercomparison
- 827 project. AGU preprint.
- 828
- Ortega-Cisneros, K., Cochrane, K., & Fulton, E. A. (2017). An Atlantis model of the southern
  Benguela upwelling system: Validation, sensitivity analysis and insights into ecosystem
  functioning. Ecological Modelling, 355, 49–63.
- 832
- Ortega-Cisneros, K., Shannon, L., Cochrane, K., Fulton, E. A., & Shin, Y.-J. (2018). Evaluating
  the specificity of ecosystem indicators to fishing in a changing environment: A model
  comparison study for the southern Benguela ecosystem. Ecological Indicators, 95, 85–98.
- 837 Ortega-Cisneros, K., Fierro Arcos, D. L., Lindmark, M., et al. (2024). An integrated global-to-
- regional scale workflow for simulating climate change impacts on marine ecosystems. ESS
- 839 Open Archive. https://doi.org/10.22541/essoar.171587234.44707846/v1
- 840

841	Payne, M. R., Barange, M., Cheung, W. W. L., MacKenzie, B. R., Batchelder, H., Cormon, X., et
842	al. (2016). Uncertainties in projecting climate-change impacts in marine ecosystems. ICES
843	Journal of Marine Science, 73(5), 1272–1282.
844	
845	Pethybridge, H. R., van Putten, I. E., Blanchard, J. L., et al. (2020). Contrasting futures for
846	Australia's fisheries stocks under IPCC RCP8.5 emissions: A multi-ecosystem model approach.
847	Frontiers in Marine Science, 7, 383.
848	
849	Petrik, C. M., Stock, C. A., Andersen, K. H., van Denderen, P. D., & Watson, J. R. (2019).
850	Bottom-up drivers of global patterns of demersal, forage, and pelagic fishes. Progress in
851	Oceanography, 176, 102124.
852	
853	Piroddi, C., Coll, M., Liquete, C., Macias Moy, D., Greer, K., Buszowski, J., Steenbeek, J.,
854	Danovaro, R., & Christensen, V. (2017). Historical changes of the Mediterranean Sea
855	ecosystem: Modelling the role and impact of primary productivity and fisheries changes over
856	time. Scientific Reports, 7.
857	
858	Polovina, J. J. (1984). Model of a coral reef ecosystem. Coral Reefs, 3, 1–11.
859	https://doi.org/10.1007/BF00306135
860	

- 861 Pozo Buil, M., Alexander, M. A., Jacox, M. G., et al. (2021). A dynamically downscaled
- 862 ensemble of future projections for the California Current System. Frontiers in Marine Science, 8,
- 863 612874. https://doi.org/10.3389/fmars.2021.612874
- 864

865	Reygondeau, G. (2019). Current and future biogeography of exploited marine groups under
866	climate change. In Cisneros-Montemayor, A. M., Cheung, W. W. L., & Ota, Y. (Eds.), Predicting
867	Future Oceans (pp. 87–101). Elsevier. https://doi.org/10.1016/B978-0-12-817945-1.00009-5
868	

- 869 Reum, J. C. P., Woodworth-Jefcoats, P., Novaglio, C., Forestier, R., Audzijonyte, A., Gardmark,
- A., et al. (2024). Temperature-dependence assumptions drive projected responses of diverse
- size-based food webs to warming. Earth's Future, 12, e2023EF003852.
- 872 https://doi.org/10.1029/2023EF003852
- 873
- Rousseau, Y., Watson, R. A., Blanchard, J. L., & Fulton, E. A. (2019). Evolution of global marine
  fishing fleets and the response of fished resources. Proceedings of the National Academy of
  Sciences, 25, 12238–12243.
- 877
- Rousseau, Y., Blanchard, J. L., Novaglio, C., et al. (2024). A database of mapped global fishing
  activity 1950–2017. Scientific Data, 11, 48. https://doi.org/10.1038/s41597-023-02824-6
- 880
- 881 Schewe, J., Gosling, S. N., Reyer, C., Zhao, F., Ciais, P., Elliott, J., Francois, L., Huber, V.,
- Lotze, H. K., Seneviratne, S. I., van Vliet, M. T. H., Vautard, R., Wada, Y., Breuer, L., Büchner,
- M., Carozza, D. A., Chang, J., Coll, M., Deryng, D., de Wit, A., Eddy, T. D., Folberth, C., Frieler,
- K., Friend, A. D., Gerten, D., Gudmundsson, L., Hanasaki, N., Ito, A., Khabarov, N., Kim, H.,
- Lawrence, P., Morfopoulos, C., Müller, C., Schmied, H. M., Orth, R., Ostberg, S., Pokhrel, Y.,
- Pugh, T. A. M., Sakurai, G., Satoh, Y., Schmid, E., Stacke, T., Steenbeek, J., Steinkamp, J.,
- Tang, Q., Tian, H., Tittensor, D., Volkholz, J., Wang, X., & Warszawski, L. (2019). State-of-the-
- 888 art global models underestimate impact from climate extremes. Nature Communications, 10,
- 889 1005.
- 890

891	Schoeman, D. S., Richardson, A. J., Poloczanska, E. S., Burrows, M. T., Duarte, C. M., &
892	Hobday, A. J. (2023). Demystifying global climate models for use in the life sciences. Trends in
893	Ecology & Evolution.
894	

- Scott, F., Blanchard, J. L., & Andersen, K. H. (2014). mizer: an R package for multispecies, traitbased and community size spectrum ecological modelling. Methods in Ecology and Evolution, 5,
  1121–1125. https://doi.org/10.1111/2041-210X.12256
- 898
- Shannon, L., Ortega Cisneros, K., Lamont, T., Winker, H., Crawford, R., Jarre, A., & Coll, M.
- 900 (2020). Exploring temporal variability in the Southern Benguela ecosystem over the past four
- 901 decades using a time-dynamic ecosystem model. Frontiers in Marine Science, 7, 540.
- 902 https://doi.org/10.3389/fmars.2020.00540
- 903
- Shin, Y.-J., & Cury, P. (2024). OSMOSE: A multispecies model to explore the impact of fishing
  on marine ecosystems.
- 906
- 907 Smith, A. D. M., Brown, C. J., Bulman, C., Fulton, E. A., Johnson, P., Kaplan, I. C., Lozano-
- Montes, H., Mackinson, S., Marzloff, M., Shannon, L. J., Shin, Y.-J., & Tam, J. (2011). Impacts
- of fishing low-trophic-level species on marine ecosystems. Science, 333(6046), 1147–1150.
- 910
- 911 Smith, A. M. D., Fulton, E. A., Day, R. W., Shannon, L. J., & Shin, Y.-J. (2015). Ecosystem
- modeling in the southern Benguela: Comparisons of Atlantis, Ecopath with Ecosim, and
- 913 OSMOSE under fishing scenarios. African Journal of Marine Science, 37, 71–83.
- 914
- 915 Stock, C. A., Alexander, M. A., Bond, N. A., Brander, K. M., Cheung, W. W. L., Curchitser, E. N.,
- Delworth, T. L., Dunne, J. P., Griffies, S. M., Haltuch, M. A., Hare, J. A., Hollowed, A. B.,

- 917 Lehodey, P., Levin, S. A., Link, J. S., Rose, K. A., Rykaczewski, R. R., Sarmiento, J. L.,
- 918 Stouffer, R. J., & Werner, F. E. (2011). On the use of IPCC-class models to assess the impact
- of climate on living marine resources. Progress in Oceanography, 88(1), 1–27.
- 920 https://doi.org/10.1016/j.pocean.2010.09.001
- 921
- Tittensor, D. P., Eddy, T. D., Lotze, H. K., & Worm, B. (2018). A protocol for the intercomparison
  of marine fishery and ecosystem models: Fish-MIP v1.0. Geoscientific Model Development, 22,
  1307–1332.
- 925
- 1926 Tittensor, D. P., Novaglio, C., Harrison, C. S., Heneghan, R. F., Barrier, N., Bianchi, D., Bopp,
- 927 L., Bryndum-Buchholz, A., Britten, G. L., Büchner, M., Cheung, W. W. L., Christensen, V., Coll,
- 928 M., Dunne, J. P., Eddy, T. D., Everett, J. D., Fernandes-Salvador, J. A., Fulton, E. A., Galbraith,
- 929 E. D., Gascuel, D., Guiet, J., John, J. G., Link, J. S., Lotze, H. K., Maury, O., Ortega-Cisneros,
- 830 K., Palacios-Abrantes, J., Petrik, C. M., du Pontavice, H., Rault, J., Richardson, A. J., Shannon,
- L., Shin, Y. J., Steenbeek, J., Stock, C. A., & Blanchard, J. (2021). Next-generation ensemble
- 932 projections reveal higher climate risks for marine ecosystems. Nature Climate Change, 11, 651–
- 933 659. https://doi.org/10.1038/s41558-021-01173-9
- 934
- 1935 Tittensor, D. P., Lotze, H. K., Eddy, T. D., Galbraith, E., Barange, M., Barrier, N., Bianchi, D.,
- Blanchard, J. L., Bopp, L., Bryndum-Buchholz, A., Carozza, D., Cheung, W. W. L., Christensen,
- 937 V., Coll, M., Jennings, S., Jones, M. C., Maury, O., Silva, T., Steenbeek, J., Verley, P., Schewe,
- J., Volkholz, J., & Büchner, M. (2023). ISIMIP2b simulation data from the global fisheries &
- marine ecosystems sector (v2.0). ISIMIP Repository. https://doi.org/10.48364/ISIMIP.780328
- 940

941	Travers, M., Shin, Y. J., Jennings, S., Machu, E., Huggett, J. A., Field, J. G., & Cury, P. M.
942	(2009). Two-way coupling versus one-way forcing of plankton and fish models to predict
943	ecosystem changes in the Benguela. Ecological Modelling, 220(21), 3089–3099.
944	
945	Travers, M., Watermeyer, K., Shannon, L. J., & Shin, YJ. (2010). Changes in food web
946	structure under scenarios of overfishing in the southern Benguela: Comparison of the Ecosim
947	and OSMOSE modelling approaches. Journal of Marine Systems, 79, 101–111.
948	https://doi.org/10.1016/j.jmarsys.2009.07.005
949	
950	Watson, R. A. (2017). A database of global marine commercial, small-scale, illegal and
951	unreported fisheries catch 1950–2014. Nature Scientific Data, 4, 170039.
952	
953	Woodworth-Jefcoats, P. A., Polovina, J. J., Dunne, J. P., & Blanchard, J. L. (2012). Ecosystem
954	size structure response to 21st-century climate projection: Large fish abundance decreases in
955	the central North Pacific and increases in the California Current. Global Change Biology, 19,
956	724–733.
957	
958	Woodworth-Jefcoats, P. A., Blanchard, J. L., & Drazen, J. C. (2019). Relative impacts of
959	simultaneous stressors on a pelagic marine ecosystem. Frontiers in Marine Science, 6, 383.
960	
961	Worm, B., Barbier, E. B., Beaumont, N., Duffy, J. E., Folke, C., Halpern, B. S., Jackson, J. B.,
962	Lotze, H. K., Micheli, F., Palumbi, S. R., Sala, E., Selkoe, K. A., Stachowicz, J. J., & Watson, R.
963	(2006). Impacts of biodiversity loss on ocean ecosystem services. Science, 314(5800), 787-
964	790.
965	
966	

Table 1. Global and regional FishMIP models with key characteristics. CMIP5 and CMIP6 indicate which models participated in each simulation round. Adapted from Tittensor et al. 2021

988 and Tittensor et al. 2018.

Model	Spatial scale	Class	CMI P5	CMI P6	Key forcing variables used	Taxonomic scope	Reference
APECOSM (Apex Predators ECOSystem Model)	Global	Composite (size- and trait-based; functional group structure)	X	X	Carbon concentrations (small phytoplankton, large phytoplankton, small zooplankton, large zooplankton), particulate organic matter (small and large), zonal and meridional currents, turbulent mixing, temperature, water density, dissolved oxygen	Sized- based communitie s (epipelagic, migratory, mesopelagi c, bathy- pelagic) and focus species	Maury (2010)

						concentration, light irradiance. All fields 3D and monthly		
Atla	intis	Regional	Composite (hybrid)	Х	X	NPP, SST, pH currents, dissolved oxygen concentration, salinity	All trophic levels and taxonomic groups can be represente d using a mix of biomass pools and age structured populations	Fulton et al. 2011
E ( T S S	BOATS Bioeconomi Marine Trophic Size- spectrum)	Global	Size-based	х	Х	Mean temperature 0–75 m, NPP	All commercial ly fished species, both finfish and invertebrate s	Carozza et al. 2016
C ( E N	DBEM Dynamic Bioclimate Invelope Model)	Global	Species distribution model	X	X	Surface and bottom O2, pH, salinity and temperature. Ice cover, current velocity, NPP, NPP pico and NPP diat. All variables on a vearly basis	956 species of exploited fishes and invertebrate s	Cheung et al. 2011
L () F N	DBPM Dynamic Benthic Pelagic Model)	Global	Composite (size- and trait-based)	х	Х	Surface and bottom temperature, phytoplankton carbon groups	All benthic and pelagic marine animals weighing between 1 mg and 1 tonne	Blanchard et al. 2012
E	EcoOcean	Global	Composite (trophodyna mic and species distribution model)	х	Х	SST, seafloor temperature, column average temperature, phytoplankton carbon groups	Includes 51 functional groups representin g the whole spectrum of marine organisms	Christense n et al. 2015; Coll et al., 2020

						from bacteria to whales, and integrates explicit information for 3 400 species of vertebrates, invertebrate s and primary	
Ecopath with Ecosim	Regional	Trophodyna mic (if Ecospace included also composite with species distribution model included)	x	x	NPP, bottom O2, SST	All trophic levels and taxonomic groups can be represente d, including age structured groups	Christense n and Walters 2004; Christense n et al., 2014
EcoTroph	Global	Trophic-level based		Х	NPP, SST, integrated mesozooplankt on carbon	Implicitly all groups, including pelagic and demersal fishes and invertebrate	Gascue, 2005; du Pontavice et al., 2021
FEISTY	Global	Composite		х	Seafloor temperature, seafloor detritus flux, mean temperature 0–100 m, integrated mesozooplankt on carbon 0–	s Small pelagic fish, large pelagic fish, demersal fish, benthic invertebrate s	Petrik et al. 2019
Macroecolo gical	Global	Size-based	Х	х	NPP, SST	Implicitly all marine organisms from 1 gram to 1 tonne	Jennings and Collingridg e (2015)
mizer	Regional	Size-based		х	Vertically integrated, size- fractionated phytoplankton and	Single plankton community, species- specific fish	Scott et al. 2014

OSMOSE	Regional	Composite (size- and trait-based)	Х		zooplankton carbon, ocean temperature SST, SSS, NPP, phytoplankton and zooplankton concentration	Fish and invert species and functional groups	Shin and Cury 2004, Travers et al. 2009
ZooMSS	Global	Composite (size- and trait-based; functional group structure)	x	x	Chlorophyll-a, SST	Flagellates, cilliates, omnivorous copepods, carnivorous copepods, larvaceans, salps, chaetognat hs, euphausiid s, jellyfish, fish	Heneghan et al. 2020

Table 2. Regions and regional marine ecosystem models (MEMs) investigated in this study with key characteristics. Forcing variables are: intpp - net primary organic carbon production by all types of phytoplankton; thethao - sea water potential temperature; phydiat - concentration of diatoms expressed as carbon in sea water; phydiaz - concentration of diazotrophs expressed as carbon in sea water; phypico-vint – mole concentration of picophytoplankton expressed as carbon in sea water; zmeso-vint - concentration of mesozooplankton expressed as carbon in sea water; zmicro-vint - concentration of microzooplankton expressed as carbon in sea water, tos – sea surface temperature.

1000
------

Region	Model	Domain Area (km2)	Spatial resolution	Forcing variables for regional models	Bias correction method applied	Reference
Baltic Sea	Ecopath with Ecosim	240 000	No	intpp, tos, bottom O2	Delta	Niiranen et al. 2013
Cook Strait, New Zealand	Ecopath with Ecosim	54	No	intpp	Delta	Eddy et al. 2014
East Bass Strait, Australia	Ecopath with Ecosim	30 623	No	intpp	Delta	Bulman et al. 2006
Eastern Bering Sea	mizer	493 506	No	phydiat-vint; phydiaz-vint; phypico-vint; zmeso-vint; zmicro-vint; tos	Delta	Reum et al. 2020

Central North Pacific	therMizer	19 694 991	No	phydiat-vint; phydiaz-vint; phypico-vint; zmeso-vint; zmicro-vint: tos	Delta	Woodwort h-Jefcoats et al. 2019
Humboldt Current	OSMOSE	4 949 170	Yes	intpp, phydiat, zmeso-vint, zmicro-vint, tos	Statistical downscalin g	Oliveros- Ramos et al. 2017
Mediterranean Sea	Ecopath with Ecosim	2 500 000	Yes	intpp; thethao	-	Piroddi et al. 2017
North Sea	Ecopath with Ecosim	570 000	No	intpp	Delta	Mackinson and Daskalov 2007
SE Australia	Atlantis	3 000 000	Yes	thetao, O2, intpp		Fulton et al. 2014
Southern Benguela	Atlantis; Ecopath with Ecosim	220 000	Yes; no	intpp; thetao	Delta	Ortega Cisneros et al. 2017 Shannon et al. 2020

1007Table 3. Summary of results. Mean percent sea surface temperature (SST) and net primary1008productivity (NPP) change from 2090-2099 relative to 1990-1999 for GFDL and IPSL Earth1009system models in CMIP5 and CMIP6.

Region	CMIP5				CMIP6				
J.	SST		NPP		S	SST		NPP	
	GFDL	IPSL	GFDL	IPSL	GFDL	IPSL	GFDL	IPSL	
Baltic Sea	53.7	100.3	-6.7	-17.1					
Cook Strait,									
New Zealand	14.8	41.9	8.5	5.1	26.9	26.3	-9.5	9.0	
East Bass									
Strait, Australia	14.4	39.6	17.8	-25.2	18.1	43.4	-8.7	23.2	
East Bering									
Sea					135.1	164.4	18.8	23.9	
Hawaii					11.9	19.4	-16.9	11.0	
Humboldt									
Current	9.0	15.5	4.4	-1.2					
Mediterranean									
Sea	13.8	26.0	23.8	-19.1					
North Sea	24.4	64.1	-6.4	-52.9					
SE Australia	12.8	24.2	3.5	-2.0					
Southern									
Benguela					11.3	12.6	3.5	18.9	
Average	20.4	44.5	6.4	-16.1	40.7	53.2	-2.5	17.2	

Table 4. Summary of results. Mean percent total consumer biomass change from 2090-2099
relative to 1990-1999 for regional and global models forced by GFDL and IPSL Earth system
models in CMIP5 and CMIP6. Note that SE Australia regional Atlantis model only ran
simulations until 2050 and results for SE Australia are shown for 2040-2049 relative to 1990-

1042 1999.

Region	CMIP5				CMIP6			
-	GFDL		IPSL		GFDL		IPSL	
	regional	global	regional	global	regional	global	regional	global
Baltic Sea	-38.9	-22.3	-63.0	-40.3				
Cook Strait,								
New Zealand	44.4	-5.0	12.3	-13.3	-60.0	-49.6	23.5	-25.9
East Bass								
Strait, Australia	2.3	15.3	-8.0	-19.8	-2.9	-31.8	15.0	-27.7
East Bering								
Sea					-68.3	-9.4	-74.0	-35.3
Hawaii					-2.3	-36.7	-2.8	-28.7
Humboldt								
Current	0.8	-4.7	-4.2	-25.5				
Mediterranean								
Sea	51.3	12.1	-21.4	-38.8				
North Sea	-7.7	-23.5	-78.0	-63.8				
SE Australia	18.3	-4.1	6.4	-13.2				
Southern								
Benguela								
Atlantis					-21.0	-12.3	-49.1	-12.4
Southern								
Benguela EwE					1.1	-12.3	19.7	-12.4
Average	10.1	-4.6	-22.3	-30.7	-25.6	-28.0	-11.3	-26.0



- Figure 1. Location of FishMIP regional model domains that completed CMIP5, CMIP6, and both
   CMIP5 & CMIP6 simulations. From left to right, regions are: Central North Pacific, Eastern
   Bering Sea, Humboldt Current, North Sea, Mediterranean Sea, Baltic Sea, Southern Benguela,
- 1059 South East Australia, East Bass Strait, and Cook Strait.



Figure 2. CMIP5 Earth system model SST and NPP forcing variables for GFDL (red) and IPSL(blue) by region for the RCP 8.5 scenario.





Figure 3. CMIP6 Earth system model SST and NPP forcing variables for GFDL (red) and IPSL(blue) by region for the SSP5-8.5 scenario.





Figure 4. CMIP5 simulations for global (red) and regional (blue) models showing percent change in total consumer biomass relative to 1990-1999 for GFDL and IPSL under no-fishing scenarios and RCP 8.5. Shaded areas indicate standard deviation for the global model ensemble. There is one regional model for each domain. Note that the regional model in SE Australia only ran simulations until 2050.





Figure 5. CMIP6 simulations for global and regional models showing percent change in total consumer biomass relative to 1990-1999 for GFDL and IPSL under no-fishing scenarios and SSP5-8.5. Shaded areas indicate standard deviation for the global model ensemble. The Southern Benguela region has two regional models; the mean of the two models is shown. Other regions have one regional model.



Figure 6. CMIP5 simulations for global and regional models showing percent change in total consumer biomass relative to 1990-1999 for GFDL and IPSL under no-fishing scenarios and RCP8.5. Note that the regional model in SE Australia only ran simulations until 2050.



Figure 7. CMIP6 simulations for global and regional models showing percent change in total
consumer biomass relative to 1990-1999 for GFDL and IPSL under no-fishing scenarios and
RCP8.5.



Figure 8. Relationship between change in total consumer biomass (delta tcb) and change in seasurface temperature (delta SST) by model.



1123 Figure 9. Relationship between change in total consumer biomass (delta tcb) and change in net

1124 primary productivity (delta NPP) by model.